Author's response to reviews

Title: Predictors of web-based follow-up response in the Prevention of Low Back Pain the Military Trial

Authors:

John D Childs (john@texpts.com)
Deydre S Teyhen (dteyhen@sbcglobal.net)
Joshua J Van Wyngaarden (josh.vanwyngaarden@gmail.com)
Brett F Dougherty (bbdougherty@gmail.com)
Bryan J Ladislas (bladislas@yahoo.com)
Gary L Helton (ghelton10@gmail.com)
Michael E Robinson (merobin@ufl.edu)
Samuel S Wu (samwu@biostat.ufl.edu)
Steven Z George (szgeorge@phhp.ufl.edu)

Version: 2 Date: 18 October 2010

Author's response to reviews: see over
2 March 2010

Guy G. Simoneau, PT, PhD, ATC
Editor-in-Chief
Journal of Orthopaedic & Sports Physical Therapy
1111 North Fairfax Street
Suite 100
Alexandria, VA 22314

Dr. Simoneau:

Enclosed is an original research report entitled “Predictors of web-based follow-up response in the Prevention of Low Back Pain the Military Trial” which I am submitting for publication in BMC Public Health. All correspondence should be directed to the primary author, Dr. John Childs.

All authors have read and provided approval for the content presented in the manuscript and for its submission to BMC Public Health. The opinions or assertions contained herein are the private views of the authors and are not to be construed as official or as reflecting the views of the U.S. Air Force, U.S. Army, or Department of Defense.

This study was funded through the Congressionally Directed Peer Reviewed Medical Research Program (#W81XWH-06-1-0564). There has been no financial affiliation or involvement with any commercial organization that has a direct financial interest in any matter included in this manuscript.

I may be reached by phone at (210) 364-7410 or by email at childsjd@gmail.com if you have any questions or need additional information. Thank you for consideration of this manuscript for publication.

John D. Childs, PT, PhD, MBA
Lt Col, USAF, BSC
Associate Professor & Director of Research
US Army-Baylor University Doctoral Program in Physical Therapy